Author's response to reviews

Title: The Feasibility, Acceptability, and Effects of an Interprofessional Nurse-Led Mental Health Promotion Intervention for Older Home Care Clients with Depressive Symptoms: A Prospective Study

Authors:

Maureen Markle-Reid (mreid@mcmaster.ca)
Carrie McAiney (mcaineyc@mcmaster.ca)
Dorothy Forbes (dorothy.forbes@ualberta.ca)
Lehana Thabane (thabanl@mcmaster.ca)
Maggie Gibson (Maggie.Gibson@sjhc.london.on.ca)
Gina Browne (browneg@mcmaster.ca)
Jeffrey Hoch (jeffrey.hoch@utoronto.ca)
Thomas Peirce (tom.peirce@hnhb.ccac-ont.ca)
Barbara Busing (barbara.busing@hnhb.ccac-ont.ca)

Version: 2  Date: 24 January 2014

Author's response to reviews: see over
January 24, 2014

Editorial Team
BMC Geriatrics

Dear BMC Editorial Team:

Thank you for your insightful comments regarding our manuscript titled, “The Feasibility, Acceptability, and Effects of an Interprofessional Nurse-Led Mental Health Promotion Intervention for Older Home Care Clients with Depressive Symptoms: A Prospective Study”, for consideration of being published in BMC Geriatrics. We have made the following changes to this paper:

- Shortened the title of the manuscript to: An Interprofessional Nurse-Led Mental Health Promotion Intervention for Older Home Care Clients with Depressive Symptoms
- Revised and reduced each section of the manuscript by 35% to enhance the clarity of the manuscript
- Reduced the number of tables from 5 to 2.

We submit the paper as an original article. It has not been published and is not under consideration for publication elsewhere. The manuscript has been read and approved by all the authors. The authors accept full responsibility for the design and conduct of the study, and all have controlled the decision to publish. The corresponding author for the article is given in the manuscript and below in this letter. The authors declare that they have no competing interests. We hope you will select this research article for publication in your journal.

Yours sincerely,

Dr. Maureen Markle-Reid, RN, MScN, Ph.D.
Associate Professor and Canada Research Chair in Aging, Chronic Disease and Health Promotion Interventions, School of Nursing
Scientific Director, Aging, Community and Health Research Unit
Associate Member, Department of Clinical Epidemiology and Biostatistics
McMaster University
1200 Main Street West, HSC 3N25B
Hamilton, Ontario, Canada L8N 3Z5
Tel: (905) 525-9140, ext. 22306; Fax: (905) 524-5199
E-mail: mreid@mcmaster.ca